Author's response to reviews

Title: Changing the obesogenic environment of severe mentally ill residential patients: ELIPS, a cluster randomised study design.

Authors:

Anne Looijmans (a.looijmans@umcg.nl)
Frederike Jörg (F.Jorg@umcg.nl)
Robert A Schoevers (R.A.Schoevers@umcg.nl)
Richard Bruggeman (r.bruggeman@umcg.nl)
Ronald P Stolk (R.P.Stolk@umcg.nl)
Eva Corpeleijn (E.Corpeleijn@umcg.nl)

Version: 2 Date: 2 October 2014

Author's response to reviews:

Revisions:

Changes made in the updated manuscript version compared to the manuscript version updated September 17, 2014, are:

- Trial Registration is adjusted: trialregistration.nl is renamed Nederlands Trialregister (Dutch trial register).
- Date of registration is added.